Author's response to reviews

Title: Associations Between Variants on ADIPOQ and ADIPOR1 with Colorectal Cancer Risk: a Chinese Case-control Study and updated Meta-analysis

Authors:

Yiyi Ou (ouyiyi2007@hotmail.com)
Peizhan Chen (pzchen@sibs.ac.cn)
Ziyuan Zhou (ziyuanzhou@yahoo.com)
Chenglin Li (clli01@sibs.ac.cn)
Jinyi Liu (jinyiliutmmu@163.com)
Kazuo Tajima (ktajima@aichi-cc.jp)
Junsheng Guo (guojs@shhyy.com)
Jia Cao (13883088490@126.com)
Hui Wang (huiwang@sibs.ac.cn)

Version: 5 Date: 2 December 2014

Author's response to reviews: see over
December 2, 14

Dr. Sergi Castellvi-Bel. Hospital Clinic, Centre Esther Koplowitz, Spain

RE: Associations Between Variants on ADIPOQ and ADIPOR1 with Colorectal Cancer Risk: a Chinese Case-control Study and updated Meta-analysis

Dear Dr. Sergi,

Thank you for your letter dated on December 2, 14 regarding the review of our aforementioned manuscript. We are glad to hear that our reports could be accepted for publication on BMC Medical Genetics and only a minor revision need to be conducted according to the requirements. According the suggestions, we have revised our manuscript and hope these could meet the criteria. The following are our responses to the comments and suggestions.

1. Please rename the Introduction section 'Background'.

Responses: We have revised the words in the manuscript.

2. Please move the funding details to the Acknowledgements section.

Responses: We have moved the funding details to the Acknowledgments section in the manuscript.

3. Tables: please combine your table captions with their tables and place them after the figure legends.
Responses: We have revised the manuscript accordingly.

Thank you for your considering, and we hope the responses will meet your and the journal criteria.

Please send all correspondences to me at the following address:

**Hui Wang, M.D., Ph.D.**

Professor, Institute for Nutritional Sciences
Shanghai Institutes for Biological Sciences
Chinese Academy of Sciences
Rm 2204, The New Biology Science Building A#
320 Yueyang Road
Shanghai, 200031
People’s Republic of China

E-mail: huiwang@sibs.ac.cn
+86-21-54920941 (Phone)
+86-21-54920291 (Fax)

Sincerely,

**Hui Wang, M.D., Ph.D.**

Professor, INS, SIBS, CAS